**Author's response to reviews**

**Title:** Radical surgery versus standard surgery for primary cytoreduction of bulky stage IIIC and IV ovarian cancer: an observational study

**Authors:**

Yulan Ren (yulan_ren@yahoo.com)
Rong Jiang (rong_junzhu@yahoo.com.cn)
Sheng Yin (yinshengyinsheng@126.com)
You Chao (youchao8888@aliyun.com)
Xi Chen (cheng_xi1@sina.com)
Jie Tang (zhangtang125@126.com)
Rongyu Zang (ryzang@yahoo.com)

**Version:** 2  
**Date:** 2 December 2014

**Author's response to reviews:**

1. We revised our manuscript to include line and page numbers as the editor's require.
2. We revised our abstract formatted according to the guidelines.
3. We included all authors' email addresses on the title page of our manuscript.